An orally administered drug prevents selection for antibiotic-resistant bacteria in the gut during daptomycin therapy



ORIGINAL

439



Valerie J. Morley (),^{1,2,*} Derek G. Sim,¹ Aline Penkevich,³ Robert J. Woods ()³ and Andrew F. Read ()^{1,4,5}

¹Department of Biology, Center for Infectious Disease Dynamics, The Pennsylvania State University, University Park, PA, USA; ²Research and Development Team, Nature's Toolbox Inc. (NTx), Rio Rancho, NM, USA; ³Division of Infectious Diseases, Department of Internal Medicine, University of Michigan, Ann Arbor, MI, USA; ⁴Department of Entomology, The Pennsylvania State University, University Park, PA, USA and ⁵Huck Institutes for the Life Sciences, The Pennsylvania State University, University Park, PA, USA

*Corresponding author. Nature's Toolbox Inc. (NTx), 7701 Innovation Way, Rio Rancho, NM 87144, USA. E-mail: valerie.morley@ntxbio.com

Received 18 May 2022; revised version accepted 16 August 2022

ABSTRACT

Background and objectives: Previously, we showed proof-of-concept in a mouse model that oral administration of cholestyramine prevented enrichment of daptomycin-resistant *Enterococcus faecium* in the gastrointestinal (GI) tract during daptomycin therapy. Cholestyramine binds daptomycin in the gut, which removes daptomycin selection pressure and so prevents the enrichment of resistant clones. Here, we investigated two open questions related to this approach: (i) can cholestyramine prevent the enrichment of diverse daptomycin mutations emerging *de novo* in the gut? and (ii) how does the timing of cholestyramine administration impact its ability to suppress resistance?

Methodology: Mice with GI *E. faecium* were treated with daptomycin with or without cholestyramine, and *E. faecium* was cultured from feces to measure changes in daptomycin susceptibility. A subset of clones was sequenced to investigate the genomic basis of daptomycin resistance.

Results: Cholestyramine prevented the enrichment of diverse resistance mutations that emerged *de novo* in daptomycin-treated mice. Whole-genome sequencing revealed that resistance emerged through multiple genetic pathways, with most candidate resistance mutations observed in the *clsA* gene. In addition, we observed that cholestyramine was most effective when administration started prior to the first dose of daptomycin. However, beginning cholestyramine after the first daptomycin dose reduced the frequency of resistant *E. faecium* compared to not using cholestyramine at all.

© The Author(s) 2022. Published by Oxford University Press on behalf of the Foundation for Evolution, Medicine, and Public Health. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Conclusions and implications: Cholestyramine prevented the enrichment of diverse daptomycin-resistance mutations in intestinal *E. faecium* populations during daptomycin treatment, and it is a promising tool for managing the transmission of daptomycin-resistant *E. faecium*.

Lay Summary: Intravenous antibiotics can select for drug-resistant bacteria in the gut. Here, we demonstrate a strategy for preventing this selection in a mouse model. We show that an orally administered drug can capture antibiotic in the gut, thus preventing selection for resistance. This strategy allows for antibiotic treatment while reducing transmission of resistance.

KEYWORDS: daptomycin; Enterococcus faecium; antimicrobial resistance; antimicrobial stewardship

BACKGROUND AND OBJECTIVES

As the antimicrobial resistance threat grows, it is critical to find ways to use antimicrobials in healthcare without fueling the spread of resistance. With this goal in mind, we recently showed proof-of-concept for a strategy to reduce the transmission of daptomycin-resistant Enterococcus faecium during daptomycin therapy [1]. Daptomycin is administered intravenously to treat blood and soft tissue infections and is one of the few antibiotics indicated for the treatment of vancomycin-resistant E. faecium (VRE) [2]. During daptomycin therapy, most daptomycin is renally excreted, but 5-10% of the dose enters the intestines [3]. This means E. faecium colonizing the intestines are exposed to daptomycin, which can select for resistance in intestinal populations [1, 4]. This off-target daptomycin selection could be an important source of hospital-transmitted daptomycin-resistant VRE because E. faecium transmission is fecal-oral [5]. We hypothesized that inactivating daptomycin locally in the intestines could prevent the enrichment and transmission of daptomycin-resistant E. faecium, while still allowing daptomycin therapy to effectively target infections in sites like the bloodstream. We tested the potential of the FDA-approved drug cholestyramine to be repurposed as a daptomycin antagonist to inactivate daptomycin in the gut. In a previous paper, we used a mouse model to show that pairing oral cholestyramine with systemic daptomycin treatment dramatically reduced the fecal shedding of daptomycin-resistant *E. faecium* [1]. Cholestyramine treatment is therefore a promising strategy that allows for daptomycin therapy while blocking the enrichment and fecal shedding of daptomycin-resistant E. faecium.

Our previous paper provided proof-of-concept that cholestyramine adjuvant therapy could limit the spread of daptomycin resistance. That work left two major open questions that we address here in the same mouse model. First, can cholestyramine prevent selection for resistance mutations that arise *de novo* in the gut during treatment? Our previous paper showed that cholestyramine successfully prevented the enrichment of daptomycin-resistant clones that were experimentally inoculated at low levels in intestinal populations. Preventing *de novo* resistance poses additional challenges, as there may be larger mutational inputs from larger sensitive populations, and a greater diversity of resistance genotypes and phenotypes may need to be suppressed. Second, how does the timing of cholestyramine administration affect resistance emergence? In our previous experiments, we began administering cholestyramine to mice 1 day prior to their first daptomycin injection. In clinical settings, it might not be practical to start cholestyramine treatment prior to the first dose of daptomycin.

METHODOLOGY

Experimental overview

We conducted two experiments. The first experiment tested the ability of cholestyramine to prevent selection for resistance mutations that occurred *de novo* during treatment. We treated 180 mice colonized with daptomycin-susceptible *E. faecium* with either 50 or 100 mg/kg subcutaneous daptomycin daily for 5 days. Half the mice were provided with a diet supplemented with cholestyramine starting 1 day prior to the first daptomycin injection; the other half were provided the same diet but without cholestyramine. We sequenced a subset of the resistant clones that emerged to determine likely resistance mechanisms.

The second experiment tested whether cholestyramine effectively inhibited the enrichment of resistance if cholestyramine treatment and daptomycin treatment began at the same time. Groups of 10 mice were inoculated orally with a mixture of daptomycin-susceptible and daptomycin-resistant E. faecium (95% susceptible, 5% resistant). This design drastically decreases the number of mice needed compared to measuring de novo resistance emergence because it eliminates the wait time for resistance mutations in the E. faecium population and focuses on selection for the resistant clone during daptomycin treatment. Mice were treated with 50 mg/kg daptomycin for 5 days. To test the effect of cholestyramine therapy timing, mice received either no cholestyramine, a cholestyraminesupplemented diet starting 1 day prior to daptomycin treatment ('early'), or a cholestyramine-supplemented diet starting shortly after their first daptomycin dose ('late').

In both experiments, the enrichment of resistance was the key readout. As in our earlier paper [1], this was measured as the percent of *E. faecium* in fecal samples that were daptomycin

resistant, and the total density of resistant *E. faecium* in feces (number per gram).

Mice and bacterial strains

Mice in all experiments were female Swiss Webster (CFW) obtained from Charles River Labs. This is an outbred mouse line. Mice were fed a standard diet (5001 Laboratory Rodent Diet) or a standard diet supplemented with 2% w/w cholestyramine resin (Sigma-Aldrich catalog # C4650). To minimize bacterial cross-contamination between animals, all mice were housed individually during experiments, and experimenters changed gloves between handling different mice.

We used the daptomycin-susceptible *E. faecium* strain BL00239-1 [MIC_c = 3.1 μ g/ml (Minimum Inhibitory Concentration computed, see below)], a vancomycin-resistant strain originally isolated from a bloodstream infection at the University of Michigan Hospital. In the experiment testing the timing of cholestyramine administration, we also used the daptomycin-resistant strain BL00239-1-R (MIC_c = 8.6 μ g/ml). This resistant strain was isolated from a mouse colonized with BL00239-1 after experimental daptomycin treatment, as previously described [1].

Experimental detail

Animal experiments were approved by the Institutional Animal Care and Use Committee at the Pennsylvania State University. All mice were pretreated with ampicillin (0.5 g/l in drinking water) for 7 days before E. faecium inoculation. Ampicillin disrupts the natural gut microbiota and facilitates Enterococcus colonization [6]. Mice that were co-housed during ampicillin pretreatment were evenly allocated among experimental treatment groups. E. faecium strains were plated from glycerol stocks and then grown overnight in liquid culture in Brain Heart Infusion broth. Mice were inoculated via oral gavage with 10⁸ CFU E. faecium suspended in saline. The same bacterial culture was used to inoculate all mice. E. faecium inoculum counts were confirmed by plating. Following E. faecium inoculation, mice were split into individual cages with untreated water and experimental diets. Daptomycin doses were administered daily starting 1 day post-inoculation via subcutaneous injection. Daptomycin doses were based on an average mouse weight for each experiment. For mice receiving cholestyramine, a cholestyramine supplemented diet (2% w/w) was provided to mice starting 1 day prior to the first daptomycin dose or starting the same day as the first daptomycin dose. Mice were allowed to eat ad libitum, and once the cholestyramine was provided, the mice continued to be maintained on this diet for the duration of the experiment. For stool collection, mice were placed in clean plastic cups, and fresh stool was collected using a sterile toothpick. Stool

samples were suspended in PBS (25 μl PBS/mg stool) and frozen with glycerol at –80°C for subsequent analysis.

Analysis of E. faecium in stool samples

Enterococcus faecium were enumerated by plating diluted fecal suspensions on selective plates (Enterococcosel agar supplemented with 16 μ g/ml vancomycin). Plates were incubated at 35°C for 40–48 h, and colonies were counted. To quantify the proportion of these bacteria that were daptomycin resistant, fecal suspensions were plated on calcium-supplemented Enterococcosel plates with 16 μ g/ml vancomycin and 10 μ g/ml daptomycin. Plates were incubated at 35°C for 40–48 h, and colonies were counted. Serially diluted fecal suspensions were each plated once on plates without daptomycin and once on plates containing daptomycin to estimate the proportion of daptomycin-resistant bacteria. The limit of detection was 20 CFU per 10 mg of feces.

In Experiment 1, a subset of *E. faecium* clones were isolated from fecal samples and analyzed by broth microdilution. Clones were purified by streaking twice on Enterococcosel agar with 16 µg/ml vancomycin and were then stored in glycerol stocks at -80°C. Broth microdilutions were performed according to Clinical & Laboratory Standards Institute guidelines [7]. After incubation, cell densities were measured by OD600 absorbance in a plate reader. OD values were fitted to a Hill function to determine the computed MIC (MIC_c), the concentration at which the Hill function crossed a cutoff two standard deviations above the mean of the negative control wells, as described previously [4]. MIC_c measurements were performed in duplicate [median variance between replicate log2 (MIC_c) measurements = 0.08].

Genome sequencing of E. faecium isolates

In Experiment 1, daptomycin resistance was observed in >5% of the *E. faecium* population in 22 of 180 mice using culture-based methods. We isolated 5 clones from each of these 22 mice for susceptibility testing. From this set of isolates, four clones from each mouse were selected for whole-genome sequencing (89 total isolates sequenced including the ancestral BL00239-1 clone). The clone selection process for sequencing was as follows: from each mouse, we chose the clones with the highest and lowest MIC_cs , and then two additional clones were selected randomly from each mouse using a random number generator.

Sequencing libraries were prepared from whole genomic DNA using the Collibri PCR-free ES DNA Library Prep Kit for Illumina with UD Indexes (sets A-D 1-96). Libraries were submitted to the University of Michigan sequencing core for paired-end 150-bp sequencing on the Illumina NovaSeq 6000. Sequencing reads were trimmed using Trimmomatic [8], and MultiQC was used to assess data quality [9]. One isolate (M3.5-5) was excluded from downstream analysis due to poor sequence quality. Trimmed reads were mapped against the BL00239-1 reference genome (generated previously [1]) and the *E. faecium* DO reference genome (NC_017960.1) using Burrows–Wheeler Aligner [10], and candidate variants were identified with The Genome Analysis Toolkit [11]. Candidate variants were annotated using SnpEff [12]. Reads from the ancestral clone were aligned to the reference genome (aligned to self or DO) to generate a list of background variants; these background variants were filtered out during variant calling. Remaining candidate variants were screened by visual inspection of alignments in Tablet [13].

Statistical analysis

Statistical analyses were run in R v1.2.1335 [14] using the packages 'nlme' [15] and 'glmmTMB' [16]. To analyze proportions of resistant bacteria, samples were plated on agar with and without daptomycin, resulting in a count of resistant bacteria and a count of total bacteria. Due to sampling, these proportions were not bounded by one, so proportion data were normalized by dividing each value by the maximum value in the data set. Proportions of resistant bacteria were analyzed using mixed binomial regression models. Absolute densities of VRE were analyzed using mixed models with an autoregressive error structure as previously described [17]. In Experiment 2, a total number of bacteria or resistant bacteria shed through time were estimated from the area under the density through time curves. Full model structures and output are shown in Supplementary File S1. Figures were created with ggplot2 [18].

RESULTS

Cholestyramine prevented the enrichment of diverse daptomycin-resistance mutations

Fecal samples were collected at Day 1 (prior to daptomycin) and Day 8 (after daptomycin treatment), and the fecal densities of total *E. faecium* and daptomycin-resistant *E. faecium* were determined (Supplementary Fig. S1). For mice with no detectable *E. faecium* in the Day 8 fecal sample (n = 19), an additional sample was collected at Day 14.

Daptomycin-resistant mutants were detected in 48% (86/180) of all mice and were detected in similar numbers or mice in both groups (cholestyramine: 37/90, no cholestyramine, 49/90; Pearson's chi-square test P = 0.10). However, the frequency of daptomycin resistance in the *E. faecium* population within a mouse never rose above 1/100 CFUs in any of the 90 mice treated with cholestyramine. In contrast, it rose above 1/100 frequency in 28% (25/90) of mice treated with daptomycin in the absence of cholestyramine, and often came to dominate

(Fig. 1). Resistance was significantly less likely to rise above 1/100 in cholestyramine-treated mice (Pearson's chi-square test P < 0.001). Thus, cholestyramine successfully prevented *de novo* emergence of daptomycin resistance in gastrointestinal (GI) *E. faecium* populations.

Cholestyramine prevented the enrichment of resistance that emerged through multiple genetic pathways

To confirm the emergence of resistance, we determined the daptomycin MIC_c for a subset of *E. faecium* isolates. For the 22 mice which had resistance >5/100 CFUs at Day 8 or Day 14, we isolated five *E. faecium* clones from each mouse's fecal sample and estimated MIC_c by broth microdilution. The broth microdilution assay confirmed that at least 16 mice harbored *E. faecium* isolates with elevated daptomycin MIC_c relative to the ancestral clone (Fig. 2).

To investigate the genetic mechanism of resistance in these isolates, we conducted whole-genome sequencing on 4 clones from each of these 22 mice (89 isolates total including the ancestral clone). A full list of observed mutations can be found in Supplementary Table S1. Across the sequenced isolates, we found mutations in several genes known to be involved in daptomycin resistance (Fig. 2). The most common candidate-resistance mutations were found in the *clsA* gene, which has previously been associated with daptomycin resistance in E. faecium [19-23]. The clsA gene encodes cardiolipin synthase, which produces cardiolipin. Increased cardiolipin content in the bacterial membrane inhibits membrane permeabilization by daptomycin [19]. We found *clsA* mutations in isolates from 14 of the 22 mice. Six different clsA mutations were observed, all of which were nonsynonymous. The most common *clsA* mutation was A20D, followed by R211L. Within 4 of the 14 mice with clsA mutations, we observed two with different mutations in different clones from the same fecal sample, and in one mouse, we observed three different *clsA* mutations. This means that multiple clsA mutations appeared independently within these populations.

In two mice, we also observed mutations in *liaS* (1301T) and *liaX* (A381T), another set of genes associated with daptomycin resistance [21, 23, 24]. The three-component LiaFSR system regulates the cellular envelope stress response, which impacts daptomycin susceptibility [25]. The three-gene cluster *liaXYZ* is hypothesized to be regulated by *liaR* [24]. In addition, we observed mutations in an HD domain containing protein previously associated with daptomycin resistance (S42Y) [21] and frameshift in the candidate-resistance gene *pspC* [21].

In summary, daptomycin resistance emerged through multiple genetic mechanisms, with strong convergence at the gene level (*clsA*). Cholestyramine prevented the enrichment of resistance in all treated mice, despite the availability of multiple genetic pathways to resistance in this system.



Figure 1. Cholestyramine inhibits enrichment of daptomycin resistance mutations in gut *E. faecium* populations. (A) Experimental design. Mice colonized with daptomycin-susceptible *E. faecium* were supplied with a standard diet or a diet supplemented with 2% cholestyramine by weight (N = 90 per diet treatment). Mice then received subcutaneous daptomycin for 5 days at 50 or 100 mg/kg daily. Fecal samples were collected at Day 8, and the frequency of daptomycin-resistant *E. faecium* in these fecal samples was determined by culture. For mice with low *E. faecium* shedding at Day 8 (n = 19), an additional sample was collected at Day 14. Numbers show how many mice had >1/100 daptomycin-resistant *E. faecium* in a Day 8 or Day 14 fecal sample. (B) Proportion of daptomycin-resistant *E. faecium* at Day 8. Fecal samples were plated on Enterococcal agar plates (Ent –DAP) and on Enterococcal plates supplemented with daptomycin (Ent +DAP). The proportion of daptomycin-resistant *E. faecium* was estimated as (CFU on Ent +DAP)/(CFU on Ent –DAP). Each point represents a fecal sample from one mouse. The dotted line shows 1/100 CFUs resistant, as in (A)



Figure 2. Daptomycin resistance in *E. faecium* clones isolated from mouse fecal samples. A total of 22 mice were identified as having *E. faecium* populations with increased daptomycin resistance based on an initial plating assay (>5/100 CFUs resistant at Day 8 or Day 14). Five *E. faecium* clones were isolated from each of these fecal samples. Daptomycin MIC_cs were determined by broth microdilution for each clone (see Methodology). Each point shows the MIC_c of a single clone (mean of two measurements). The line shows the MIC_c of the ancestral clone BL00239-1 (mean of four measurements) with 95% CI (dotted lines). Daptomycin doses are indicated at the top of each panel. A subset of these isolates was sequenced, and colors indicate mutations in genes associated with daptomycin resistance. The isolates marked with an asterisk had two candidate resistance mutations in *clsA* and an HD-domain containing protein

Early cholestyramine administration better prevented the enrichment of resistance

Both early and late cholestyramine treatments significantly reduced the proportion of *E. faecium* that were daptomycin-

resistant relative to no-cholestyramine controls (Fig. 3A; mixed binomial model, early \times day P=0.006, late \times day P=0.03, Model 1 in Supplementary File S1). 'Early' cholestyramine had a substantially larger effect size at all time points (Fig. 3A, Model



Figure 3. Effect of timing of cholestyramine administration on enrichment for resistance. Mice treated with daptomycin were provided with a standard diet (no chol), cholestyramine starting 1 day prior to the first daptomycin injection (early chol), or cholestyramine starting immediately after the first daptomycin injection (late chol) (N = 10 per treatment). Fecal samples were collected at time points during and after daptomycin treatment, and densities of *E. faecium* and daptomycin-resistant *E. faecium* were determined by plating. (**A**) The proportion of *E. faecium* in mouse fecal samples that were daptomycin resistant (CFU +DAP/CFU –DAP). Means are shown, with points indicating values for individual samples. Note that fecal samples with no detectable *E. faecium* (n = 16) were excluded from this analysis. (**B**) The area under the curve of density through time (AUC) for the absolute densities of total *E. faecium* and daptomycin-resistant *E. faecium* in fecal samples over the total duration of the experiment. All mice received daptomycin, with cholestyramine administered starting at different times (none, early, late). Points indicate AUCs for individual mice for total *E. faecium* (triangles, CFU measured on Enterococcosel plates with daptomycin). Means and SE are shown (N = 10)

1 in Supplementary File S1). Only the 'early' treatment significantly reduced total shedding of daptomycin-resistant *E. faecium*, as measured by area under the curve of density through time (AUC) (Fig. 3B; generalized linear model (glm), early P = 0.01, late P = 0.12, Model 2 in Supplementary File S1). 'Early' cholestyramine reduced absolute shedding of resistant *E. faecium* (AUC) by 92% compared to no cholestyramine.

In daptomycin-treated mice, cholestyramine treatment did not affect the total shedding of *E. faecium* (Fig. 3B; glm, early P=0.86, late P=0.88, Model 3 in Supplementary File S1). In control mice that received no daptomycin but received identical cholestyramine treatments, cholestyramine alone also did not affect total *E. faecium* shedding (Supplementary Fig. S2; glm, early P=0.19, late P=0.22, Model 4 in Supplementary File S1).

CONCLUSIONS AND IMPLICATIONS

Here, we have shown that cholestyramine can prevent enrichment of resistance following the *de novo* appearance of daptomycin resistance in *E. faecium* populations colonizing the intestines (Experiment 1, Fig. 1). Previously, we had shown that cholestyramine could prevent the enrichment of preexisting resistant clones [1] (a result we saw again in Experiment 2; Fig. 3). These results further show the promise of cholestyramine treatment for preventing off-target resistance evolution emerging via multiple genetic pathways during daptomycin treatment. Ideally, cholestyramine would allow clinicians to effectively treat bloodstream and soft tissue infections with daptomycin, while preventing the emergence of daptomycin-resistant *E. faecium* from intestinal populations and risk of onward transmission.

Daptomycin resistance in E. faecium arises through chromosomal mutations and there are many possible genetic pathways to increased resistance [21]. This means that within a patient, especially in large E. faecium populations colonizing the gut, there is a high probability that daptomycin-resistance mutations will appear through random mutation. Treatment then imposes selection, which can enrich for these resistant clones. This process of resistance emergence has been observed in hospital patients—E. faecium isolated from patient perirectal swabs is more resistant following daptomycin treatment, and genomic data suggest that this resistance arises *de novo* within patients [4]. Here, we observe the same process in a mouse model. We detected the presence of de novo resistance (at least one resistant clone) in 48% of mice (Fig. 1). In the absence of cholestyramine, daptomycin treatment resulted in the up-selection of this resistance above 1/100 CFUs in about half of those mice (Fig. 1). Resistance emerged so frequently likely because mutations can confer resistance via multiple genetic pathways (Fig. 2). Should similar rates of spontaneous resistance mutation and emergence occur in patients colonized with *E. faecium*, adjuvant therapies that can inhibit daptomycin activity in the GI tract could have substantial impact. Importantly, cholestyramine effectively suppressed resistance emergence even though resistance was readily accessible through multiple genetic pathways (Figs 1 and 2). We hypothesize that cholestyramine works by eliminating selection pressure for daptomycin resistance, thus preventing the enrichment of resistant clones, independent of the genetic mechanism of resistance.

We note that mice in these experiments were inoculated with large doses of bacteria (10⁸ CFU). Under these conditions, daptomycin-resistance mutations may already have been present in the initial inoculum. This is one possible explanation for the observation of the same resistance mutations in different mice—these mutations could have been present in the initial culture. Regardless of whether these resistance mutations first appeared in the initial culture or within the mice, the important finding here is that cholestyramine prevented selection for these mutations.

These results also show that the timing of cholestyramine administration may matter. We found in mice that cholestyramine was most effective at preventing resistance when administration began 24 h before the first daptomycin dose, although beginning cholestyramine after the first daptomycin dose still had some suppressive effect on the frequency of resistance. An essential step in translating cholestyramine treatment to human subjects will be understanding the pharmacokinetics of cholestyramine and daptomycin, especially when the drugs reach the intestines after administration and in what quantities. This pharmacokinetic data should inform decisions about the optimal cholestyramine regimens.

Cholestyramine is one of several adjunctive therapies in development that could prevent off-target resistance emergence in the gut during antibiotic treatment [26-29]. These treatments rely on antibiotic antagonists, which bind to or inactivate antibiotics locally in the intestines. Daptomycin is an especially attractive antibiotic target, because daptomycin resistance emerges easily and often in the gut through point mutations, as shown here in our mouse model and in patient data [4]. In addition, cholestyramine is an attractive drug to repurpose as an adjuvant, because it is an FDA-approved drug that has been in use with minimal side effects for over 50 years [30, 31], and it could inexpensively be repurposed to prevent the spread of resistance. These adjunctive therapies are exciting new tools for antimicrobial stewardship, which offer the possibility of using antibiotics to treat patients without the onward transmission of drug-resistant pathogens.

SUPPLEMENTARY DATA

Supplementary data are available at EMPH online.

ACKNOWLEDGMENTS

We thank Kevin Tracy for assistance in generating the nanopore sequencing and analysis pipeline.

FUNDING

The Pennsylvania State University (to A.F.R.) and the National Institutes of Health (R01 AI143852) (to R.J.W.). The funders had no role in study design, data collection and interpretation, or the decision to submit the work for publication.

Conflict of interest: None declared.

DATA AVAILABILITY

Data are available on Dryad (DOI https://doi.org/10.5061/dryad. rxwdbrvbm). Raw sequencing reads are available through NCBI Bioproject PRJNA832698.

REFERENCES

- 1. Morley VJ, Kinnear CL, Sim DG *et al*. An adjunctive therapy administered with an antibiotic prevents enrichment of antibiotic-resistant clones of a colonizing opportunistic pathogen. *eLife* 2020;**9**:e58147.
- Isenman H, Fisher D. Advances in prevention and treatment of vancomycin-resistant *Enterococcus* infection. *Curr Opin Infect Dis* 2016; 29:577–82.
- 3. Woodworth JR, Nyhart EH Jr, Brier GL et al. Single-dose pharmacokinetics and antibacterial activity of daptomycin, a new lipopeptide

antibiotic, in healthy volunteers. *Antimicrob Agents Chemother* 1992;**36**: 318–25.

- Kinnear CL, Hansen E, Morley VJ et al. Daptomycin treatment impacts resistance in off-target populations of vancomycin-resistant Enterococcus faecium. PLoS Biol 2020;18:e3000987.
- 5. Arias CA, Murray BE. The rise of the *Enterococcus*: beyond vancomycin resistance. *Nat Rev Microbiol* 2012;**10**:266–78.
- Ubeda C, Taur Y, Jenq RR et al. Vancomycin-resistant Enterococcus domination of intestinal microbiota is enabled by antibiotic treatment in mice and precedes bloodstream invasion in humans. J Clin Invest 2010;120:4332–41.
- CLSI. Performance Standards for Antimicrobial Susceptibility Testing. 31st edn. Wayne, PA: Clinical and Laboratory Standards Institute, 2021.
- Bolger AM, Lohse M, Usadel B. Trimmomatic: a flexible trimmer for Illumina sequence data. *Bioinformatics* 2014;30:2114–20.
- Ewels P, Magnusson M, Lundin S *et al.* MultiQC: summarize analysis results for multiple tools and samples in a single report. *Bioinformatics* 2016;**32**:3047–8.
- Li H, Durbin R. Fast and accurate short read alignment with Burrows– Wheeler transform. *Bioinformatics* 2009;25:1754–60.
- Van der Auwera G, O'Connor B. Genomics in the Cloud: Using Docker, GATK, and WDL in Terra. 1st edn. Sebastopol, California: O'Reilly Media, 2020.
- Cingolani P, Platts A, Wang LL *et al*. A program for annotating and predicting the effects of single nucleotide polymorphisms, SnpEff: SNPs in the genome of *Drosophila melanogaster* strain w1118; iso-2; iso-3. *Fly* (*Austin*) 2012;**6**:80–92.
- Milne I, Bayer M, Cardle L et al. Tablet—next generation sequence assembly visualization. Bioinformatics 2010;26:401-2.
- 14. R Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria: R Foundation for Statistical Computing, 2021.
- Pinheiro J, Bates D, DebRoy S et al. nlme: Linear and Nonlinear Mixed Effects Models. R package version 3.1-159, 2021. https://CRAN.R-pro ject.org/package=nlme
- Brooks ME, Kristensen K, van Benthem KJ *et al.* glmmTMB balances speed and flexibility among packages for zero-inflated generalized linear mixed modeling. *R J* 2017;**9**:378–400.
- Pollitt LC, Reece SE, Mideo N et al. The problem of auto-correlation in parasitology. PLoS Pathog 2012;8:e1002590.

- Wickham H. Ggplot2: Elegant Graphics for Data Analysis. New York: Springer-Verlag, 2009.
- Zhang T, Muraih JK, Tishbi N *et al.* Cardiolipin prevents membrane translocation and permeabilization by daptomycin. *J Biol Chem* 2014; 289:11584–91.
- Davlieva M, Zhang W, Arias CA *et al.* Biochemical characterization of cardiolipin synthase mutations associated with daptomycin resistance in *Enterococci. Antimicrob Agents Chemother* 2013;57:289–96.
- Diaz L, Tran TT, Munita JM et al. Whole-genome analyses of Enterococcus faecium isolates with diverse daptomycin MICs. Antimicrob Agents Chemother 2014;58:4527–34.
- Lellek H, Franke GC, Ruckert C et al. Emergence of daptomycin non-susceptibility in colonizing vancomycin-resistant Enterococcus faecium isolates during daptomycin therapy. Int J Med Microbiol 2015;305:902–9.
- Arias CA, Panesso D, McGrath DM *et al.* Genetic basis for in vivo daptomycin resistance in Enterococci. N Engl J Med 2011;365:892–900.
- 24. Khan A, Davlieva M, Panesso D *et al.* A novel mechanism of resistance against antimicrobial peptides involves extracellular sensing coupled with membrane remodeling mediated by a single protein. *Open Forum Infect Dis* 2017;4:S133.
- Munita JM, Panesso D, Diaz L et al. Correlation between mutations in liaFSR of Enterococcus faecium and MIC of daptomycin: revisiting daptomycin breakpoints. Antimicrob Agents Chemother 2012;56:4354–9.
- Rooney CM, Ahmed S, Wilcox MH. Protecting the microbiota. J Infect Dis 2021;223:S290-5.
- 27. Kokai-Kun JF, Roberts T, Coughlin O et al. Use of ribaxamase (SYN-004), a β-lactamase, to prevent Clostridium difficile infection in β-lactam-treated patients: a double-blind, phase 2b, randomised placebo-controlled trial. *Lancet Infect Dis* 2019;**19**:487–96.
- 28. Tarkkanen A-M, Heinonen T, Jõgi R et al. P1A recombinant beta-lactamase prevents emergence of antimicrobial resistance in gut microflora of healthy subjects during intravenous administration of ampicillin. Antimicrob Agents Chemother 2009;53:2455–62.
- 29. de Gunzburg J, Ghozlane A, Ducher A *et al*. Protection of the human gut microbiome from antibiotics. *J Infect Dis* 2018;**217**:628–36.
- Jacobson TA, Armani A, McKenney JM *et al.* Safety considerations with gastrointestinally active lipid-lowering drugs. *Am J Cardiol* 2007;99: 47C–55C.
- Beckett R, Wilhite A. Drugs that affect lipid metabolism. In: Ray S (ed). Side Effects of Drugs Annual. Vol 37. Amsterdam: Elsevier, 2015, 559–65.